Author's response to reviews

Title: Pharmacology as a foreign language: A preliminary evaluation of podcasting as a supplementary learning tool for non-medical prescribing students

Authors:

Oonagh Meade (lwxom2@nottingham.ac.uk)
Bowskill Dianne (dianne.bowskill@nottingham.ac.uk)
Joanne S Lymn (joanne.lymn@nottingham.ac.uk)

Version: 3 Date: 9 December 2009

Author's response to reviews:

Details of consent have been added to the methods section. Page 10, paragraph 2..

Methodology has been changed to Methods, page 8.

Visible vertical lines have been removed from tables.

Figures have been cropped.